\begin{thebibliography}{10}

\bibitem{treadmarks96}
{\sc Amza, C., Cox, A., Dwarkadas, S., Hyams, C., Li, Z., and Zwaenepoel, W.}
\newblock Treadmarks: Shared memory computing on networks of workstations.
\newblock {\em IEEE Computer 29}, 2 (Feb. 1996), 18--28.

\bibitem{pooma95}
{\sc Atlas, S., Banerjee, S., Cummings, J., Hinker, P., Srikant, M., Reynders,
  J., and Tholburn, M.}
\newblock {POOMA}: A high performance distributed simulation environment for
  scientific applications.
\newblock In {\em Proceedings of Supercomputing '95\/} (Washington, D.C., Nov.
  1995).

\bibitem{superquake96}
{\sc Bao, H., Bielak, J., Ghattas, O., O'Hallaron, D., Kallivokas, L.,
  Shewchuk, J., and Xu, J.}
\newblock Earthquake ground motion modeling on parallel computers.
\newblock In {\em Proc. Supercomputing '96\/} (Pittsburgh, PA, Nov. 1996).
\newblock See also www.cs.cmu.edu/$\sim$quake/.

\bibitem{quakejournal97}
{\sc Bao, H., Bielak, J., Ghattas, O., O'Hallaron, D., Kallivokas, L.,
  Shewchuk, J., and Xu, J.}
\newblock Large-scale simulation of elastic wave propagation in heterogeneous
  media on parallel computers.
\newblock {\em Computer Methods in Applied Mechanics and Engineering\/} (1997).
\newblock To appear.

\bibitem{barnard92}
{\sc Barnard, S., and Simon, H.}
\newblock A fast multilevel implementation of recursive spectral bisection for
  partitioning unstructured problems.
\newblock Tech. Rep. RNR-92-033, NASA Ames Research Center, Nov. 1992.

\bibitem{shrimp94}
{\sc Blumrich, M., Li, K., Alpert, K., Dubnicki, C., Felten, E., and Sandberg,
  J.}
\newblock Virtual memory mapped network interface for the {SHRIMP}
  multicomputer.
\newblock In {\em Proc. 21th Intl. Symp. on Computer Architecture\/} (May
  1994), ACM, pp.~142--153.

\bibitem{dongarra94}
{\sc Dongarra, J., Lumsdaine, A., Pozo, R., and Remington, K.}
\newblock A sparse matrix library in {C++} for high performance architectures.
\newblock In {\em Proceedings of the Second Object Oriented Numerics
  Conference\/} (1994), pp.~214--218.

\bibitem{duff89}
{\sc Duff, I.~S., Grimes, R.~G., and Lewis, J.~G.}
\newblock Sparse matrix test problems.
\newblock {\em {ACM} Transactions on Mathematical Software 15}, 1 (Mar. 1989),
  1--14.

\bibitem{farhat88}
{\sc Farhat, C.}
\newblock A simple and efficient automatic {FEM} domain decomposer.
\newblock {\em Comp. \& Struct. 28}, 5 (1988), 579--602.

\bibitem{GiMiTe95}
{\sc Gilbert, J., Miller, G., and Teng, S.-H.}
\newblock Geometric mesh partitioning: Implementation and experiments.
\newblock In {\em 9th International Parallel Processing Symposium\/} (Santa
  Barbara, CA, Apr. 1995), IEEE, pp.~418--427.

\bibitem{gropp96}
{\sc Gropp, W., and Smith, B.}
\newblock The design of data-structure-neutral libraries for the iterative
  solution of sparse linear systems.
\newblock {\em Scientific Programming 5\/} (1996), 329--336.

\bibitem{chacospectral}
{\sc Hendrickson, B., and Leland, R.}
\newblock An improved spectral graph partitioning algorithm for mapping
  parallel computations.
\newblock {\em SIAM J.\ Sci.\ Comput. 16}, 2 (1995), 452--469.

\bibitem{hinrichs94}
{\sc Hinrichs, S., Kosak, C., O'Hallaron, D., Stricker, T., and Take, R.}
\newblock An architecture for optimal all-to-all personalized communication.
\newblock In {\em Proc. SPAA '94\/} (Cape May, NJ, June 1994), ACM,
  pp.~310--319.

\bibitem{hockney96}
{\sc Hockney, R.}
\newblock {\em The Science of Computer Benchmarking}.
\newblock SIAM, Philadelphia, PA, 1996.

\bibitem{dash92}
{\sc Lenoski, D., Laudon, J., Gharachorloo, K., Weber, W., Gupta, A., Hennessy,
  J., Horowitz, M., and Lam, M.}
\newblock The {S}tanford {DASH} multiprocessor.
\newblock {\em IEEE Computer 25}, 3 (Mar. 1992), 63--79.

\bibitem{magistrale96}
{\sc Magistrale, H., McLaughlin, K., and Day, S.}
\newblock A geology-based 3-{D} velocity model of the {L}os {A}ngeles basin
  sediments.
\newblock {\em Bulletin of the Seismological Society of America 86\/} (1996),
  1161--1166.

\bibitem{mttv93}
{\sc Miller, G.~L., Teng, S.-H., Thurston, W., and Vavasis, S.~A.}
\newblock Automatic {M}esh {P}artitioning.
\newblock In {\em Graph Theory and Sparse Matrix Computation}, A.~George, J.~R.
  Gilbert, and J.~W.~H. Liu, Eds. Springer-Verlag, New York, 1993.

\bibitem{mpisuper}
{\sc {MPI Forum}}.
\newblock {MPI: A Message Passing Interface}.
\newblock In {\em Proc. Supercomputing '93\/} (Portland, OR, Nov. 1993),
  ACM/IEEE, pp.~878--883.

\bibitem{sfprops96}
{\sc O'Hallaron, D., and Shewchuk, J.}
\newblock Properties of a family of parallel finite element simulations.
\newblock Tech. Rep. CMU-CS-96-141, School of Computer Science, Carnegie Mellon
  University, Dec. 1996.

\bibitem{pothen92}
{\sc Pothen, A., Simon, H., and Liou, K.}
\newblock Partitioning sparse matrices with eigenvectors of graphs.
\newblock {\em SIAM Journal on Matrix Analysis and Applications 11\/} (1990),
  430--452.

\bibitem{schwabe92}
{\sc Schwabe, E., Blelloch, G., Feldmann, A., Ghattas, O., Gilbert, J., Miller,
  G., O'Hallaron, D., Shewchuk, J., and Teng, S.}
\newblock A separator-based framework for automated partitioning and mapping of
  parallel algorithms for numerical solution of {PDE}s.
\newblock In {\em Proc.1992 DAGS/PC Symposium\/} (June 1992), pp.~48--62.

\bibitem{t3esync}
{\sc Scott, S.}
\newblock Synchronization and communication in the {T3E} multiprocessor.
\newblock In {\em Proc. 7th. Intl. Conf. on Arch. Support for Prog. Lang. and
  Oper. Systems\/} (Boston, MA, Oct. 1996), ACM, pp.~26--36.

\bibitem{simon91}
{\sc Simon, H.}
\newblock Partitioning of unstructured problems for parallel processing.
\newblock {\em Comput. Sys. Engr. 2}, 3 (1991), 135--148.

\bibitem{ics-decouple}
{\sc Stricker, T., Stichnoth, J., O'Hallaron, D., Hinrichs, S., and Gross, T.}
\newblock Decoupling synchronization and data transfer in message passing
  systems of parallel computers.
\newblock In {\em Proc. of 9th Intl. Conference on Supercomputing\/}
  (Barcelona, Spain, July 1995), ACM, pp.~1--10.

\bibitem{zhou97}
{\sc Zhou, Y., Iftode, L., Singh, J., Li, K., Toonen, B., Schoinas, I., Hill,
  M., and Wood, D.}
\newblock Relaxed consistency and coherence granularity in {DSM} systems: A
  performance evaluation.
\newblock In {\em Proceedings of the SIGPLAN Symposium on Principles and
  Practice of Parallel Programming (PPoPP)\/} (Las Vegas, NV, June 1997), ACM,
  pp.~193--205.

\end{thebibliography}

\newcommand{\comment}[1]{}

\documentstyle[11pt,fullpage,twoside,./cmu-titlepage,times,./epsf,./fancyheadings]{article}

\setcounter{topnumber}{4}
\renewcommand{\topfraction}{1}
\setcounter{bottomnumber}{4}
\renewcommand{\bottomfraction}{1}
\setcounter{totalnumber}{6}
\renewcommand{\textfraction}{0}
\renewcommand{\floatpagefraction}{0.8}

% \renewcommand{\baselinestretch}{0.93}

% \setlength{\topsep}{0pt}
% \setlength{\partopsep}{0pt}
% \setlength{\itemsep}{0pt}

%\def\headrule{{\if@fancyplain\headrulewidth\plainheadrulewidth\fi
%  \hrule\@height\headrulewidth\@width\headwidth}}

\raggedbottom
\pagestyle{plain}

\title{Spark98: Sparse Matrix Kernels for\\
Shared Memory and Message Passing Systems}

\author{David R. O'Hallaron}

%\date{}

\trnumber{CMU-CS-97-178}

\support{ 
Effort sponsored in part by the Advanced Research Projects Agency and
Rome Laboratory, Air Force Materiel Command, USAF, under agreement
number F30602-96-1-0287, in part by the National Science Foundation
under Grant CMS-9318163, and in part by a grant from the Intel
Corporation. The U.S. Government is authorized to reproduce and
distribute reprints for Governmental purposes notwithstanding any
copyright annotation thereon.  The views and conclusions contained
herein are those of the authors and should not be interpreted as
necessarily representing the official policies or endorsements, either
expressed or implied, of the Advanced Research Projects Agency, Rome
Laboratory, or the U.S. Government. Author's email address: droh@cs.cmu.edu
}

\keywords{benchmark, finite element, parallel computing, sparse matrix}

\abstract{\small 
Spark98 is a collection of sparse matrix kernels for shared memory and
message passing systems.  Our aim is to provide system builders with a
set of example sparse matrix codes that are simple, realistic, and
portable.  Each kernel performs a sequence of sparse matrix vector
product operations using matrices that are derived from a family of
three-dimensional finite element earthquake applications.  We describe
the computational structure of the kernels, summarize their
performance on a parallel system, and discuss some of the insights
that such kernels can provide.  In particular we notice that efficient
parallel programming of sparse codes requires careful partitioning of
data references, regardless of the underlying memory system.  So on
one hand, efficient shared memory programs can be just as difficult to
write as efficient message passing programs.  On the other hand,
shared memory programs are not necessarily less efficient than message
passing programs.  }

\begin{document}
\bibliographystyle{acm}

\maketitle

% Print page numbers for table of contents using small roman numbers.
\renewcommand{\thepage}{\roman{page}}
\tableofcontents
\newpage
%\listoffigures
%\newpage

\setlength{\parskip}{\medskipamount}

\section*{About this Report}

The Spark98 kernels (including an electronic copy of this report) can 
be obtained from the Web page\\
at www.cs.cmu.edu/\verb.~.quake/spark98.html.

\cleardoublepage

% Revert to arabic numbers, and also reset counter to page 1.
\renewcommand{\thepage}{\arabic{page}}
\setcounter{page}{1}

%\topmargin -14.4pt
\pagestyle{fancy}
\addtolength{\headheight}{2.4pt}
\renewcommand{\sectionmark}[1]{\markboth{}{#1}}
\renewcommand{\subsectionmark}[1]{}
\lhead[\rm\thepage]{}
\chead[{\sl David R. O'Hallaron}]
{{\sl \rightmark}}
\rhead[]{\rm\thepage}
\cfoot[]{}
\thispagestyle{empty}

\renewcommand{\thefootnote}{\arabic{footnote}}

\section{Introduction}

The multiplication of a sparse matrix by a dense vector is central to
many computer applications, including scheduling applications based on
linear programming and applications that simulate physical systems.
For example, the Quake project at Carnegie Mellon University uses a
sequence of more than 16,000 sparse matrix-vector product (SMVP)
operations to simulate the motion of the ground during the first 40
seconds of an aftershock of the 1994 Northridge earthquake in the San
Fernando Valley~\cite{quakejournal97,superquake96}. The sparse matrix
consists of over 13 million rows and 180 million nonzero entries,
where each nonzero entry is a dense $3 \times 3$ submatrix of double
precision floating point numbers. Repeated executions of a single SMVP
routine account for over 70\% of the simulation's computation time.

A sparse computation like the SMVP is fundamentally different from a
dense matrix computation because its memory reference pattern is
irregular and depends on the nonzero structure of the sparse matrix,
which in turn depends on the physics of the underlying system and the
goals of the simulation. For example, the number and distribution of
mesh nodes in an earthquake simulation is determined by both the
density of the soil and the frequencies of interest to the user.

The data-dependent nature of sparse computations presents difficulties
for system builders who need access to sparse codes in order to
evaluate design options. A kernel must be simple enough that designers
can understand it without having to wade through masses of obscure
code. If the code is simple then it can be modified, and thus becomes
a useful tool for experimentation. A kernel must also be realistic in
that it represents a larger class of applications. Finally, a kernel
must also be portable so that it can be run without modification on a
large number of platforms.

Existing sparse matrix collections like the Harwell-Boeing
suite~\cite{duff89} are important tools for designers of numerical
algorithms and partitioning algorithms. And existing software
libraries that support sparse matrix computations are useful tools for
application programmers~\cite{pooma95,dongarra94,gropp96}. However, to
date the designers of parallel systems have not had access to a set of
public domain sparse matrix kernels that are simple, realistic, and
portable.  Towards that end, we introduce the Spark98 kernels, a set
of 10 SMVP kernels for shared memory and message passing
systems.\footnote{Spark is a loose acronym for ``sparse kernels'', but
actually the name reflects our hope that the kernels will spark
interest in sparse computations among system designers.}  We expect
that builders of cache coherent shared memory
multiprocessors~\cite{dash92,shrimp94}, distributed memory
systems~\cite{t3esync}, message passing libraries~\cite{mpisuper}, and
distributed shared memory libraries~\cite{treadmarks96,zhou97} will
find that the Spark98 kernels are useful tools for understanding the
performance of irregular codes on their systems. In our own experience
with the Spark98 kernels we notice that efficient parallel programming
of sparse codes requires careful partitioning of data references,
regardless of the underlying memory system. Because equal care must be
given to the partitioning of references, we find that for sparse
codes, efficient shared memory programs are just as difficult to write
as efficient message passing programs. On the other hand, shared
memory programs are not necessarily less efficient than message
passing programs.

Each Spark98 kernel consists of a C program and a partitioned finite
element mesh, which is described by an input file. There are five
programs ({\sc smv}, {\sc lmv}, {\sc rmv}, {\sc mmv}, and {\sc hmv})
and two meshes (sf10 and sf5), for a total of ten kernels, which are
summarized in Figure~\ref{fig:spark98}. In the figure, $n$ is the
number of nodes, $nz$ is the number of nonzero entries in the
corresponding sparse matrix (where each entry is a $3 \times 3$
submatrix of double precision floating point numbers), and $p$ is the
number of processing elements (PEs). (We use the term {\em processing
element} instead of the more common term {\em node} to avoid confusion
with the nodes of finite element meshes.)  For reference, all of the
symbols used in this paper are summarized in Figure~\ref{fig:symbols}.

Each Spark98 program consists of roughly $1,000$ lines of code, most
of which serve to parse the input files; the actual SMVP code is quite
small. Kernels such as {\sc lmv} and {\sc rmv} use trivial
partitioning algorithms that are computed at runtime, and thus can run
on an arbitrary number of PEs, as indicated in the rightmost column of
Figure~\ref{fig:spark98}. Other kernels like {\sc mmv} and {\sc hmv}
rely on a partition that is precomputed offline and stored in a
file. For these kernels we provide precomputed partitions for $1, 2,
4, 8, \ldots, 64, 128$ PEs.

\begin{figure}
\begin{center}
\small
\begin{tabular}{|l||l|l|}
\hline
\multicolumn{3}{|c|}{Kernels based on mesh sf10 ($n=7,294$, $nz=97,138$)}\\ 
\hline
\multicolumn{1}{|c||}{Kernel name} & 
\multicolumn{1}{c|}{Description} & 
\multicolumn{1}{c|}{Number of PEs ($p$)}\\ 
\hline
\hline
{\sc smv}/sf10 & sequential&$p = 1$\\ 			
{\sc lmv}/sf10 & parallel, shared memory using locks&$p \geq 1$\\		
{\sc rmv}/sf10 & parallel, shared memory using reductions&$p \geq 1$\\
{\sc mmv}/sf10 & parallel, message passing&$p = 1, 2, 4, 8, \ldots, 128$\\ 	
{\sc hmv}/sf10 & parallel, hybrid shared memory and message passing&
$p = 1, 2, 4, 8, \ldots, 128$\\
\hline
\multicolumn{3}{l}{}\\ 
\multicolumn{3}{l}{}\\ 
\hline
\multicolumn{3}{|c|}{Kernels based on mesh sf5 ($n=30,169$, $nz=410,923$)}\\ 
\hline
\multicolumn{1}{|c||}{Kernel name} & 
\multicolumn{1}{c|}{Description} & 
\multicolumn{1}{c|}{Number of PEs ($p$)}\\ 
\hline
\hline
{\sc smv}/sf5 & sequential&$p = 1$\\ 			
{\sc lmv}/sf5 & parallel, shared memory using locks&$p \geq 1$\\		
{\sc rmv}/sf5 & parallel, shared memory using reductions&$p \geq 1$\\
{\sc mmv}/sf5 & parallel, message passing&$p = 1, 2, 4, 8, \ldots, 128$\\ 	
{\sc hmv}/sf5 & parallel, hybrid shared memory and message passing&
$p = 1, 2, 4, 8, \ldots, 128$\\
\hline
\end{tabular}
\end{center}
\caption{The Spark98 kernels.}
\label{fig:spark98}
\end{figure}

The Spark98 meshes are called sf10 and sf5. They were developed by the
Quake project at Carnegie Mellon to model earthquake-induced ground
motion in the San Fernando Valley of Southern California. Each mesh is
an unstructured three-dimensional finite element model of the earth
underneath the San Fernando Valley. Sf10 is a small-sized mesh
($n=7,294$, $nz=97,138$) and sf5 is moderate-sized ($n=30,169$,
$nz=410,923$).
\begin{figure}
\small
\begin{center}
\begin{tabular}{|l||l|}
\hline
\multicolumn{1}{|c||}{Symbol} & 
\multicolumn{1}{c|}{Description}\\ 
\hline
\hline
$p$	&number of subdomains or PEs\\
$L$	&number of locks\\
\hline
$n$ 	&nodes in mesh (and rows in matrix)\\
$e$ 	&edges in mesh (including self-edges)\\
$nz$  	&nonzero $3\times3$ submatrices, assuming nonsymmetric storage.\\
\hline
$n_i$ 	&nodes in mesh (and rows in matrix) on subdomain $i$\\
$e_i$ 	&edges in subdomain $i$ (including self-edges)\\
$nz_i$ 	&nonzero $3\times3$ submatrices in the matrix on subdomain $i$, 
assuming nonsymmetric storage\\
\hline
\end{tabular}
\end{center}
\caption{Summary of symbols used in this paper.}
\label{fig:symbols}
\end{figure}

The Spark98 programs are written in ANSI C and consist of a baseline
sequential program ({\sc smv}) and four parallel programs ({\sc lmv},
{\sc rmv}, {\sc mmv}, and {\sc hmv}).  Each program computes a
sequence of SMVP pairs, $w = Kv$ followed by $w' = K'v'$, where $K$
and $K'$ are symmetric $n \times n$ sparse matrices with identical
nonzero structures, and $v$, $v'$, $w$, and $w'$ are $n \times 1$
dense vectors.  Each vector entry is a $3 \times 1$ subvector of
doubles.  Each of the $nz$ nonzero matrix entries is a $3 \times 3$
submatrix of doubles. In order to conserve precious memory, matrices
are stored in a compressed sparse row (CSR) format where only the
submatrices along the diagonal and in the upper triangle are
actually stored in memory.  As we will see in
Section~\ref{sec:programs}, exploiting symmetry in this way makes the
kernels much harder to parallelize because each diagonal nonzero
submatrix contributes to two subvectors of the output vector.

{\sc Lmv} and {\sc rmv} are simple parallel shared memory programs based on locks
and reductions, respectively. Each uses a trivial partitioning algorithm
whereby each thread is assigned a contiguous set of matrix rows with
roughly the same number of nonzeros.  For {\sc lmv}, each thread updates a
shared output vector $w$, which is protected by locks. For {\sc rmv}, each
thread updates its own private copy of the output vector. These private
vectors are then summed to produce the final result vector $w$.
Because of the simple partitioning algorithm, both {\sc lmv} and {\sc rmv} can run
on an arbitrary number of processing elements (PEs). 

{\sc Mmv} is a parallel message passing program that uses a partition
precomputed by a sophisticated geometric recursive bisection mesh
partitioning algorithm~\cite{mttv93}. {\sc Hmv} is a hybrid shared
memory program that employs the same mesh partition as {\sc mmv}, but uses
shared memory copy operations rather than sends and receives to
transfer data.  Precomputed partitions are supplied for these kernels
for $1, 2, 4, \ldots, 128$ PEs. The shared memory programs can be
compiled to use either the Posix or SGI thread
interfaces. The message passing program is based on the industry
standard MPI interface~\cite{mpisuper}.

In sum, Spark98 is a set of sparse matrix kernels that are designed
for system builders who want to evaluate the performance of sparse
codes on their systems.  The kernels are small and easy to understand,
realistic in that the underlying meshes were developed by an
application group, and portable in that they are written in C using
standard shared memory and message passing interfaces.

The remainder of the paper is organized as follows.  Sections
\ref{sec:meshes} and \ref{sec:programs} describe the Spark98 meshes
and programs in more detail. Section \ref{sec:discussion} summarizes
the performance of the kernels on some parallel systems and discusses
some of the insights from using the kernels.

\section{The Spark98 meshes}
\label{sec:meshes}
The Spark98 kernels are based on a pair of three-dimensional
unstructured finite element meshes, called sf10 and sf5, that model a
volume of earth under the San Fernando Valley roughly 50 km x 50 km x
10 km in size~\cite{magistrale96}.  They are {\em unstructured} in the
sense that the neighbors of a node cannot be determined implicitly and
must be determined explicitly from adjacency information stored in
memory. Sf10 and sf5 are artifacts of the CMU Quake project, which has
developed computer techniques for predicting earthquake-induced ground
motion in large basins.  The ``sf'' in the names is an abbreviation
for San Fernando, and the digit indicates the highest frequency
seismic wave (in seconds) that the mesh is able to resolve.  For
example, sf10 resolves waves with 10 second periods and sf5 resolves
waves with 5 second periods. The sf10 mesh is illustrated in
Figure~\ref{fig:sf10mesh}.  Beverly Hills is in the lower right-hand
corner. The town of San Fernando is in the midst of the darkly shaded
region near the upper left corner.

\begin{figure}
\centerline{ 
\epsfxsize=5in 
\epsfbox{sf10.ele.all.bw.epsf} 
}
\caption{The Spark98 10-second San Fernando mesh.}
\label{fig:sf10mesh}
\end{figure}

Sf10 and sf5 are composed of different sized tetrahedra (i.e.,
pyramids with triangular bases). Each tetrahedron is called an {\em
element}, and the vertices of the tetrahedra are called {\em nodes}.
Edges are undirected, and each node has an implied self-edge.
Figure~\ref{fig:meshprops} summarizes the basic topological properties
of the Spark98 meshes and their corresponding sparse matrices. 

\begin{figure}
\begin{center}
\small
\begin{tabular}{|l||l|r|r|r|r|r|r|}
\hline
&&
\multicolumn{3}{|c|}{Mesh Entities} & 
\multicolumn{3}{|c|}{Nonzero submatrices}\\
\cline{3-5}
\cline{6-8}
Mesh&Description&
\multicolumn{1}{|c|}{Nodes ($n$)} & 
\multicolumn{1}{|c|}{Edges ($e$)} &
\multicolumn{1}{|c|}{Elements} &
total($nz$)&avg/row&max/row\\
\hline
\hline
sf10	&10-second San Fernando mesh 	&7,294  &52,216
&35,025		&97,138 & 13 & 26\\ 
sf5	&5-second San Fernando mesh	&30,169	& 220,546
& 151,239	& 410,923 & 14 & 30\\
\hline
\end{tabular}
\end{center}
\caption{Basic properties of the Spark98 meshes and their matrices.} 
\label{fig:meshprops}
\end{figure}

The Spark98 meshes are extremely sparse, with only 13-14 nonzero
submatrices per row on average and 30 nonzero submatrices per row at most. As
we will see in the next section, the number of nonzero submatrices in each row
determines the trip count of the inner loop of the basic SMVP
algorithm. Sparse rows result in short inner loops, which are hard to
pipeline or vectorize.

Partitioned versions of the Spark98 meshes are provided for $1, 2, 4,
\ldots, 64, 128$ PEs.  These partitioned meshes provide a wide range of
computation to communication ratios (i.e., the number of floating
point operations per PE divided by the number of communicated words
per PE), ranging from 500:1 for small numbers of PEs down to 50:1 for
large numbers of PEs~\cite{sfprops96}. The partitioned meshes also
provide a challenging communication pattern for the SMVP that lies
somewhere between a simple nearest neighbor pattern and a complete
exchange~\cite{hinrichs94}. See \cite{sfprops96} for a complete
characterization of the computation and communication properties of
the sf10 and sf5 partitioned meshes, along with some even larger
meshes that could appear in future versions of the Spark kernels.

\section{The Spark98 programs}
\label{sec:programs}
Each of the five Spark98 programs executes a sequence of 
SMVP pairs of the form
\scriptsize
\tt
\begin{tabbing}
ttt\=ttt\=ttt\= ttt\=ttt\= \kill 
\><read input mesh>\\
\><generate sparse matrices and vectors>\\
\>{\bf loop} i = 1 {\bf to} iters\\	
\>\>w = K * v\\
\>\>w' = K' * v'\\             	
\>{\bf endloop}\\   	
\end{tabbing}
\rm
\normalsize
where $K$ and $K'$ are sparse symmetric {\em stiffness matrices} and
$v$, $v'$, $w$, and $w'$ are dense vectors. Each iteration consists of
a pair of SMVP operations, rather than a single SMVP operation, to
approximate the worst case cache behavior of the SMVP when it is used
in a real finite element application. In a real application, SMVP
operations are separated by references to other data structures and
I/O operations which can replace the cache lines associated with the
matrix. So in the worst case, each SMVP operation in the real
application must reload its data into the cache, and alternating SMVP
operations with different sparse matrices in the kernels approximates
this behavior.

A stiffness matrix ${K}$ can be likened to an adjacency matrix of
the nodes of the mesh; ${K}$ contains a $3 \times 3$ submatrix for
each pair of nodes connected by each of the $e$ edges of the mesh (including
self-edges). If the mesh has $n$ nodes, then $K$ has dimension $3n
\times 3n$ and each vector has dimension $3n \times 1$.  (The constant
factors of 3 are due to the fact that the underlying quake simulations
model the displacement of the ground in each of three directions.)
All scalars are double-precision floating-point numbers.

A stiffness matrix consists of $nz = 2e - n$ nonzeros, where $e$ and
$n$ are the number of edges and nodes respectively in the underlying
mesh. However, because simulations are often memory bound, the matrix
coefficients are stored in a symmetric compressed sparse row (CSR)
format where only the $e$ nonzero submatrices on the diagonal and in the
upper triangle are actually stored in memory. These $e$ nonzero
submatrices are stored row-wise in an $e$-submatrix coefficient vector $A$.

The nonzero structure of the stiffness matrix is represented by two
integer vectors, $Acol$ and $Aindex$.  $Acol$ is an $e$-element vector
such that $Acol[i]$ is the column number of submatrix $A[i]$, and
$Aindex$ is an $(n+1)$-element vector such that $Aindex[i]$ is the
starting position in $A$ of row $i$.  $Aindex[n]$ marks the end of the
last row and is always set to $e+1$.  Note that while each Spark98
kernel requires two distinct coefficient vectors, one for $K$ and the
other for $K'$, it requires only a single pair of $Acol$ and $Aindex$
vectors because $K$ and $K'$ have identical sparsity structure.

Figure~\ref{fig:smvp} shows the sequential SMVP routine 
that lies at the heart of all of the Spark98 kernels. Note that
{\em local\_smvp()} actually peforms the operation $w \leftarrow w +
Kv$ rather than $w \leftarrow Kv$, because the former is easier to
implement than the latter. However, $w$ is always initially zero, so
the effect is the same.

The pattern of data references in {\em local\_smvp()} is ideal in some
ways and less than ideal in others.  On the one hand, it is ideal in
the sense that it scans sequentially through the $A$ coefficient
vector (indexed by variable {\em Anext} in line 25 of
Figure~\ref{fig:smvp}).  The sequential access pattern provides good
spatial locality and minimizes memory bank conflicts in systems with
interleaved memory systems.
\begin{figure}
\scriptsize
\tt
\begin{tabbing}
ttt\=ttt\=ttt\= ttt\=ttt\=ttt\=ttt\=\kill 
1\>{\bf void} local\_smvp({\bf int} nodes, {\bf double} (*A)[3][3], {\bf int} *Acol, {\bf int} *Aindex,\\
2\>\>\>\>\>\>{\bf double} (*v)[3], {\bf double} (*w)[3], {\bf int} firstrow, {\bf int} numrows) \{ \\
3\>\>{\bf int} i;\\
4\>\>{\bf int} Anext, Alast, col;\\
5\>\>{\bf double} sum0, sum1, sum2;\\
6\\
7\>\>{\bf for} (i = firstrow; i < (firstrow + numrows); i++) \{ \\
8\>\>\>Anext = Aindex[i];\\
9\>\>\>Alast = Aindex[i + 1];\\
10\\
11\>\>\>sum0 = A[Anext][0][0]*v[i][0] + A[Anext][0][1]*v[i][1] + A[Anext][0][2]*v[i][2];\\
12\>\>\>sum1 = A[Anext][1][0]*v[i][0] + A[Anext][1][1]*v[i][1] + A[Anext][1][2]*v[i][2];\\
13\>\>\>sum2 = A[Anext][2][0]*v[i][0] + A[Anext][2][1]*v[i][1] + A[Anext][2][2]*v[i][2];\\
14\\
15\>\>\>Anext++;\\
16\>\>\>{\bf while} (Anext < Alast) \{ \\
17\>\>\>\>col = Acol[Anext];\\
18\>\>\>\>sum0 += A[Anext][0][0]*v[col][0] + A[Anext][0][1]*v[col][1] + A[Anext][0][2]*v[col][2];\\
19\>\>\>\>sum1 += A[Anext][1][0]*v[col][0] + A[Anext][1][1]*v[col][1] + A[Anext][1][2]*v[col][2];\\
20\>\>\>\>sum2 += A[Anext][2][0]*v[col][0] + A[Anext][2][1]*v[col][1] + A[Anext][2][2]*v[col][2];\\
21\\      
22\>\>\>\>w[col][0] += A[Anext][0][0]*v[i][0] + A[Anext][1][0]*v[i][1] + A[Anext][2][0]*v[i][2];\\
23\>\>\>\>w[col][1] += A[Anext][0][1]*v[i][0] + A[Anext][1][1]*v[i][1] + A[Anext][2][1]*v[i][2];\\
24\>\>\>\>w[col][2] += A[Anext][0][2]*v[i][0] + A[Anext][1][2]*v[i][1] + A[Anext][2][2]*v[i][2];\\
25\>\>\>\>Anext++;\\
26\>\>\>\} \\
27\>\>\>w[i][0] += sum0;\\
28\>\>\>w[i][1] += sum1;\\
29\>\>\>w[i][2] += sum2;\\
30\>\>\} \\
31\>\} \\
\end{tabbing}
\rm
\normalsize
\caption{Spark98 SMVP routine. }
\label{fig:smvp}
\end{figure}

On the other hand, the access pattern is less than ideal in the
following sense.  Suppose we are manipulating some submatrix $A_{ij}$
during the course of the sequential access of $A$. Since we are using
a storage scheme that exploits symmetry, then we must be sure to use
$A_{ij}$ to record the contribution of both $A_{ji}$ and $A_{ij}$ to
the result.  In particular, $w_i \leftarrow w_i + A_{ij} v_j$ (lines
18-20, 27-29) and $w_j \leftarrow w_j + A_{ji}v_i$ (lines 22-24).  The
important implication is that some sets of references scan through $v$
and $w$ sequentially (i.e., the references to $v_i$ and $w_i$), but
other accesses are irregular (i.e., the references $v_j$ and $w_j$).
The impact of the irregular accesses of $v_j$ and $w_j$ is tempered
somewhat by the fact that each vector entry is really three numbers
(recall that the underlying simulation has three degrees of freedom)
and thus there is a limited amount of spatial locality.  Still, these
irregular accesses are a fundamental reason that the SMVP is a
challenging and interesting computational kernel.

The SMVP becomes much easier to compute if we do not exploit symmetry
and instead store the entire matrix. In this case, each row of the
matrix contributes to a unique subvector of $w$.  If we assign each thread
a disjoint set of matrix rows, then there is no contention among
either the reads of $A$ or the writes to $w$, and the only possible
contention occurs during the reads of $v$. However, experience with
applications groups makes it very clear that memory efficiency is
crucial, and thus the Spark98 kernels are based on the more
computationally interesting representation that exploits symmetry.

\subsection{{\sc Smv}: baseline sequential program}

{\sc Smv} is the baseline sequential program.
Each SMVP $w = Kv$ is performed with the following procedure calls:

\scriptsize
\tt
\begin{tabbing}
ttt\=ttt\=ttt\= ttt\=ttt\= \kill 
\>zero\_vector(w, 0, n);\\
\>local\_smvp(n, A, Acol, Aindex, v, w, 0, n); 
\end{tabbing}
\rm
\normalsize

{\em Zero\_vector(w, j, m)} zeros out subvectors $w[j], \ldots , w[j+m-1]$,
i.e., the first $m$ subvectors of $w$ starting at position $j$.

Speedup numbers for the parallel Spark98 programs should always be
reported with respect to {\sc smv}, not the single-PE version of the
parallel program.

\subsection{{\sc Lmv}: shared memory program using locks}

{\sc Lmv} is a parallel program based on a shared memory model, where a
program consists of a collection of $p$ threads that share a single
address space. The number of threads is arbitrary and can be specified
at runtime. Each thread $i$ is assigned a disjoint and contiguous set
of rows of the sparse matrix so that each set has roughly the same
number of nonzeros. Since there are on the order of thousands of rows
and only tens of nonzeros per row, it is trivial to generate good
partitions. Thread $i$ performs the SMVP on its set of rows, using the
following call sequence:

\scriptsize
\tt
\begin{tabbing}
ttt\=ttt\=ttt\= ttt\=ttt\= \kill 
\>barrier();\\
\>zero\_vector(w, firstrow[i], numrows[i]);\\
\>barrier();\\
\>local\_smvp(n, A, Acol, Aindex, v, w, firstrow[i], numrows[i]);\\ 
\end{tabbing}
\rm
\normalsize

{\em Firstrow[i]} and {\em numrows[i]} describe the set of rows
assigned to thread $i$, {\em barrier()} performs a barrier
synchronization among all of the threads.
Since different
threads might update the same subvector of $w$, a set of $L > 0$ locks
guarantees consistent updates of $w$:

\begin{minipage}{4in}
\scriptsize
\tt
\begin{tabbing}
ttt\=ttt\=ttt\= ttt\=ttt\= \kill 
\>setlock(col \% L);\\
22\>w[col][0] += A[Anext][0][0]*v[i][0] + ...\\
23\>w[col][1] += A[Anext][0][1]*v[i][0] + ...\\
24\>w[col][2] += A[Anext][0][2]*v[i][0] + ...\\
\>unsetlock(col \% L);\\
\end{tabbing}
\rm
\normalsize
\end{minipage}
\begin{minipage}{2.5in}
\scriptsize
\tt
\begin{tabbing}
ttt\=ttt\=ttt\= ttt\=ttt\= \kill 
\>\>\>setlock(i \% L);\\
\>\>27\>w[i][0] += sum0;\\
\>\>28\>w[i][1] += sum1;\\
\>\>29\>w[i][2] += sum2;\\
\>\>\>unsetlock(i \% L);\\
\end{tabbing}
\rm
\normalsize
\end{minipage}

If lock $k$ is unset, then {\em setlock(k)} atomically sets lock $k$ and
returns. Otherwise, {\em setlock(k)} waits until the lock becomes
unset, then atomically sets it.  {\em Unsetlock(k)} unsets lock $k$.  In
general, more locks means fewer synchronizations between threads, with
the best case being a unique lock for each subvector. However, most
systems limit the number of locks.  Thus the number of locks $L$ can
be specified at runtime.

It is interesting to observe that the {\em only} reason we need locks
in the first place is because we are exploiting the symmetry of the
coefficient matrix by storing only the diagonal and upper triangle.
If we were to store the entire sparse matrix, then each thread would
update a disjoint portion of the result vector, and the program would
be trivially parallelized. This is a classic space/time tradeoff that
makes the symmetric version of the SMVP much more interesting than the
usual nonsymmetric case.

\subsection{{\sc Rmv}: shared memory program using reductions}

A simple modification to the {\sc lmv} program eliminates the need for the
locks, but increases the memory requirement and introduces an extra
reduction operation. The idea is for each thread $i$ to update its own
private $w$ vector. These $p$ private vectors are then summed (reduced) to
form the final result vector $w$. 

\scriptsize
\tt
\begin{tabbing}
ttt\=ttt\=ttt\= ttt\=ttt\= \kill 
\>barrier();\\
\>zero\_vector(privatew[i], 0, n);\\
\>local\_smvp(n, A, Acol, Aindex, v, privatew[i], firstrow[i], numrows[i]);\\
\>barrier();\\
\>add\_vectors(privatew, w, firstrow[i], numrows[i]);\\ 
\end{tabbing}
\rm
\normalsize

{\em Privatew} is an array of $p$ private $w$ vectors, where are
summed by the {\em add\_vectors()} routine to produce the final result
vector $w$. The code for the {\em add\_vectors()} routine is shown in
Figure~\ref{fig:addvectors}. Notice that each thread sums the subvectors
in $n/p$ rows of $p$ private vectors, for a total of $n$ subvectors per
thread, independent of the number of threads. This suggests that {\sc rmv}
will not scale well as we increase the number of threads.

\begin{figure}
\scriptsize
\tt
\begin{tabbing}
ttt\=ttt\=ttt\= ttt\=ttt\= \kill 
1\>{\bf void} add\_vectors({\bf double} (**tmpw)[3], {\bf double} (*w)[3], {\bf int} firstrow, {\bf int} numrows) \{ \\
2\>\> {\bf int} i, j;\\
3\>\> {\bf double} sum0, sum1, sum2;\\
4\>\\  
5\>\>{\bf for} (i = firstrow; i < (firstrow+numrows); i++) \{ \\
6\>\>\>sum0 = sum1 = sum2 = 0.0;\\
7\>\>\>{\bf for} (j=0; j< threads; j++) \{ \\
8\>\>\>\> sum0 += tmpw[j][i][0];\\
9\>\>\>\> sum1 += tmpw[j][i][1];\\
10\>\>\>\> sum2 += tmpw[j][i][2];\\
11\>\>\>    \} \\
12\>\>\>w[i][0] = sum0;\\
13\>\>\>w[i][1] = sum1;\\
14\>\>\>w[i][2] = sum2;\\
15\>\>\}\\
16\>\} \\
\end{tabbing}
\rm
\normalsize
\caption{Reduction operation for {\sc rmv}. }
\label{fig:addvectors}
\end{figure}

\subsection{{\sc Mmv}: message passing program}

The {\sc mmv} program is based on a {\em message passing model} where PEs
have private memories and use standard MPI message passing primitives
to transfer data between the memories~\cite{mpisuper}. Unlike {\sc lmv} and
{\sc rmv}, which are parallelized at runtime using a trivial partitioning
algorithm, {\sc mmv} is parallelized at compile time by a sophisticated
partitioning algorithm that partitions the mesh into $p$ disjoint sets
of tetrahedral elements. Each such set is called a {\em subdomain} and
is assigned to exactly one PE (we use the terms subdomain, PE,
and thread interchangeably). The partitioner is based on a recursive
geometric bisection algorithm that divides the elements equally among
the subdomains while attempting to minimize the total number of nodes
that are shared by subdomains, and hence minimize the total
communication volume~\cite{mttv93,schwabe92}.  The algorithm
enjoys provable upper bounds on the volume of communication, and in
practice~\cite{GiMiTe95} generates partitions that are comparable to
those produced by other modern
partitioners~\cite{barnard92,farhat88,chacospectral,pothen92,simon91}.

To compute ${ w = Kv}$ on a set of PEs, we must consider the data
distribution by which vectors and matrices are stored.  The vectors
${ v}$ and ${ w}$ are stored in a distributed fashion according
to the mapping of nodes to PEs induced by the partition of elements
among PEs.  If a node $i$ resides in several PEs (because $i$ is a
vertex of several elements mapped to different PEs), the values $v_i$
and $w_i$ are replicated on those PEs. The matrix ${ K}$ is
distributed so that $K_{ij}$ resides on any PE on which nodes $i$ and
$j$ both reside. Figure~\ref{fig:cutmatrix} demonstrates this method
of distributing data. 

Given this method of distributing data, the multiplication ${ w =
Kv}$ is performed in two steps: (1) {\em Computation step:} each PE
independently computes a local matrix-vector product over the
subdomain that resides on that PE. (2) {\em Communication step:} PEs
that share nodes communicate and sum their nodal ${ w}$ values
into correct global values for each node.  In
Figure~\ref{fig:cutmatrix}, PEs 1 and 2 must exchange the values of
the interface nodes 4, 5, and 6.  This exchange-and-add process is
also referred to as {\em full assembly}.  The code for each PE has the
following form:
\begin{figure}
\centerline{\epsfbox{newcutmatrix.epsf}}
\caption{A finite element mesh and corresponding stiffness matrix ${ K}$,
         distributed among two PEs.  Xs represent nonzero $3 \times 3$
         submatrices.  Note that some nodes are shared by both PEs,
         as are some submatrices (corresponding to shared edges).}
\label{fig:cutmatrix} 
\end{figure}

\scriptsize
\tt
\begin{tabbing}
ttt\=ttt\=ttt\= ttt\=ttt\= \kill 
\>zero\_vector(w, 0, n);\\
\>local\_smvp(n, A, Acol, Aindex, v, w, 0, n);\\
\>full\_assemble(w);\\
\end{tabbing}
\rm
\normalsize

The variable $n$ on PE $i$ is the number of nodes $n_i$ assigned to PE
$i$. Similarly, the vector $A$ holds the nonzero submatrices of
sparse coefficient matrix $K$ that are assigned to PE $i$.  

All communication between PEs occurs during the {\em full\_assemble()}
routine, which resolves the values of the interface nodes using MPI
message passing primitives. {\em Full\_assemble()} has access to
communication schedules on each PE that tell it which nodes are shared
with other PEs, which PEs those nodes are shared with, and how to map
between nodes and values included in each message. 

{\em Full\_assemble()} consists of two steps. Step 1: After an initial
barrier synchronization (to ensure that the previous computation step
has finished on each PE), each PE posts an asynchronous receive
operation for each message that it expects to receive from other PEs.
Messages are maximally aggregated, and thus each PE receives at most
one message from any other PE. The asynchronous receive operations are
non-blocking, and their purpose is to tell the message passing system
where to store arriving messages in user memory. Each message is stored in
its own memory; messages do not share storage. Step 2: After another
barrier synchronization (to ensure that all PEs have posted their
receive operations), each PE sends a message to every PE with which it
shares mesh nodes. After the messages are sent, each PE waits for all
of the expected messages to arrive, and then sums the values in the
messages into the appropriate nodal values. This two-step
communication pattern, which is an example of {\em deposit model
communication}~\cite{ics-decouple}, ensures that user-level memory
exists for every message and thus avoids unnecessary copying by the
message passing system.

\subsection{{\sc Hmv}: hybrid message passing/shared memory program}

{\sc Mmv} is distinguished by the aggressive way that it partitions
its data references. {\sc Hmv} is a hybrid shared memory program that
achieves an equivalent effect by using arrays to simulate the private
local data structures. For example, the local coefficient matrix $A$
on PE $i$ in the {\sc mmv} message passing program becomes $A[i]$ in
the shared memory program. Similarly, the number of nodes $n$ on PE
$i$ becomes $n[i]$.  If we use shared memory copy operations instead
of message passing transfers to implement the full assembly phase,
then the SMVP on each thread has the following form:

\scriptsize
\tt
\begin{tabbing}
ttt\=ttt\=ttt\= ttt\=ttt\= \kill 
\> zero\_vector(w[i], 0, n[i]);\\
\>local\_smvp(n[i], A[i], Acol[i], Aindex[i], v[i], w[i], 0, n[i]);\\
\>shmem\_full\_assemble(w, i);\\
\end{tabbing}
\rm
\normalsize

{\sc Hmv} is a hybrid in the sense that it combines aggressive partitioning
of references from the message passing program with the shared memory
mechanism for the full assembly step.  

For both {\sc mmv} and {\sc hmv}, the computational load on each PE
(or thread) is proportional to the number of nonzero submatrices
assigned to each PE. If there are $e_i$ edges and $n_i$ nodes assigned
to subdomain $i$, then there are a total of $nz_i = 2e_i - n_i$
nonzero submatrices.  Since each nonzero submatrix requires a $3 \times 3$
dense matrix-vector product operation, subdomain $i$ performs a
total of $18nz_i$ floating point operations.
Figure~\ref{fig:nonzerospersubdomain} shows the distribution of
nonzeros for partitioned meshes distributed with the Spark98 kernels.
However, since symmetry is exploited in the implementation, only $e_i$
nonzero submatrices are actually stored in memory.
\begin{figure}
\begin{minipage}{3in}
\begin{center}
\small
\begin{tabular}{|c|l||r|r|}
\hline 
\multicolumn{2}{|c||}{subdomains} & 
\multicolumn{1}{c|}{sf10} & 
\multicolumn{1}{c|}{sf5}\\
\hline
\hline
1	&	&97,138	&410,923\\
\hline
	&min	&48,488	&207,201\\
2	&avg	&49,493	&208,226\\
	&max	&49,498	&209,252\\
\hline
	&min	&25,043	&104,584\\
4	&avg	&25,218	&105,522\\
	&max	&25,403	&107,135\\
\hline
	&min	&12,849	&52,609\\	
8	&avg	&13,087	&53,930\\	
	&max	&13,225	&54,740\\	
\hline
	&min	&6,608	&26,927\\	
16	&avg	&6,819	&27,604\\	
	&max	&7,108	&28,227\\	
\hline
	&min	&3,447	&13,953\\	
32	&avg	&3,610	&14,278\\	
	&max	&3,784	&14,811\\	
\hline
	&min	&1,806	&7,123\\	
64	&avg	&1,942	&7,468\\	
	&max	&2,158	&8,232\\	
\hline
	&min	&967	&3,696\\	
128	&avg	&1,053	&3,942\\	
	&max	&1,208	&4,278\\	
\hline
\end{tabular}
\end{center}
\caption{Distribution of nonzero $3 \times 3$ submatrices 
($nz_i = 2e_i - n_i$)  per subdomain, assuming nonsymmetric storage.  
Only $e_i$ nonzero submatrices are actually stored in memory.  }
\label{fig:nonzerospersubdomain}
\end{minipage}
\hspace{.4in}
\begin{minipage}{3.0in}
\begin{center}
\small
\begin{tabular}{|c|l||r|r|r|r|}
\hline 
\multicolumn{2}{|c||}{subdomains} & 
\multicolumn{1}{c|}{sf10} & 
\multicolumn{1}{c|}{sf5}\\
\hline
	&min	&4	&5	\\ 
1	&avg	&13	&13	\\
	&max	&26	&30	\\
\hline
	&min	&4	&4	\\ 
2	&avg	&13	&13	\\
	&max	&26	&30	\\
\hline
	&min	&4	&4	\\ 
4	&avg	&13	&13	\\
	&max	&26	&28	\\
\hline
	&min	&4	&4	\\
8	&avg	&13	&13	\\
	&max	&25	&27	\\
\hline
	&min	&4	&4	\\
16	&avg	&12	&12	\\
	&max	&23	&26	\\
\hline
	&min	&4	&4	\\
32	&avg	&12	&12	\\
	&max	&23	&23	\\
\hline
	&min	&4	&4	\\
64	&avg	&10	&11	\\
	&max	&19	&22	\\
\hline
	&min	&4	&4	\\
128	&avg	&9	&11	\\
	&max	&22	&24	\\
\hline
\end{tabular}
\end{center}
\caption{Distribution of nonzero submatrices per row 
for the subdomain with the fewest average submatrices per row
(assuming nonsymmetric storage). }
\label{fig:nonzerosfewest}
\end{minipage}
\end{figure}

In Figure~\ref{fig:nonzerospersubdomain}, we see that the load is
reasonably well distributed for smaller numbers of processors, but
that for larger numbers of processors, the load between two PEs can
differ by as much as 20\%.  This imbalance, which is due to the fact
that the partitioner distributes mesh elements rather than mesh edges,
highlights an interesting tradeoff.  On the one hand, if the
partitioner distributes mesh elements, then the local sparse matrix
coefficients can be generated independently on each PE using legacy
sequential code.  On the other hand, the amount of work on each
processor is more directly related to the number of nonzeros on each
processor than the number of elements, which suggests that the
partitioner should distribute mesh edges rather than mesh
elements. The Quake project opted to simplify the generation of the
local stiffness matrices by distributing mesh elements. This is the
approach we have taken with the Spark98 kernels, at the cost of
non-optimal load balancing.

Another important mesh property is the number of nonzero 
submatrices per row, which determines the length of the inner loop of the
local SMVP routine. Longer inner loops are generally more efficient
because the loop and pipeline overheads can amortized better. The
statistics for the partitioned meshes distributed with the Spark98
kernels are shown in Figure~\ref{fig:nonzerosfewest}, assuming
nonsymmetric storage. The important point about these numbers is that
they show us how sparse the matrices really are: no row in any matrix
has more than 30 nonzero submatrices, and on average has a paltry 13.

The duration of the communication phase is determined by both the
number and size of the messages transferred by each PE.  These
statistics for the partitioned meshes distributed with the Spark98
kernels are summarized in Figures~\ref{fig:msgspersubdomain} and
\ref{fig:commvolpersubdomain}. The interesting point is that the
communication phase tends to consist of a large number of fairly short
messages. Thus, message overhead matters, and we cannot count on huge
messages to amortize this cost.
\begin{figure}
\begin{minipage}{3in}
\begin{center}
\small
\begin{tabular}{|c|l||r|r|}
\hline 
\multicolumn{2}{|c||}{subdomains} & 
\multicolumn{1}{c|}{sf10} & 
\multicolumn{1}{c|}{sf5}\\
\hline
\hline
	&min	&2	&2 \\
2	&avg	&2	&2 \\	
	&max	&2	&2 \\	
\hline
	&min	&4	&2 \\
4	&avg	&5	&4 \\	
	&max	&6	&6 \\	
\hline
	&min	&4	&4 \\	
8	&avg	&8	&7 \\	
	&max	&12	&12\\	
\hline
	&min	&4	&4 \\	
16	&avg	&10	&11\\	
	&max	&18	&20\\	
\hline
	&min	&6	&6\\	
32	&avg	&16	&14\\	
	&max	&30	&30\\	
\hline
	&min	&6	&8\\	
64	&avg	&19	&18\\	
	&max	&38	&40\\	
\hline
	&min	&6	&8\\	
128	&avg	&25	&21\\	
	&max	&62	&52\\	
\hline
\end{tabular}
\end{center}
\caption{Number of messages per subdomain.}
\label{fig:msgspersubdomain}
\end{minipage}
\hspace{.4in}
\begin{minipage}{3in}
\begin{center}
\small
\begin{tabular}{|c|l||r|r|}
\hline 
\multicolumn{2}{|c||}{subdomains} & 
\multicolumn{1}{c|}{sf10} & 
\multicolumn{1}{c|}{sf5}\\
\hline
\hline
	&min	&1,716	&5,052\\
2	&avg	&1,716	&5,052\\
	&max	&1,716	&5,052\\
\hline
	&min	&1,344	&2,592\\
4	&avg	&1,839	&5,160\\
	&max	&2,352	&7,746\\
\hline
	&min	&1,152	&3,258\\
8	&avg	&1,905	&4,893\\
	&max	&2,550	&7,080\\
\hline
	&min	&942	&2,028\\
16	&avg	&1,617	&3,849\\
	&max	&2,208	&5,292\\
\hline
	&min	&750	&1,926\\
32	&avg	&1,362	&3,045\\
	&max	&2,172	&4,476\\
\hline
	&min	&492	&1,026\\
64	&avg	&1,110	&2,382\\
	&max	&1,764	&4,296\\
\hline
	&min	&396	&762\\
128	&avg	&885	&1,797\\
	&max	&1,740	&3,360\\
\hline
\end{tabular}
\end{center}
\caption{Communication words per subdomain.}
\label{fig:commvolpersubdomain}
\end{minipage}
\end{figure}

\section{Discussion}
\label{sec:discussion}
The Spark98 programs are points on a continuum that is defined by how
well each program partitions its data references among computational
threads.  At one end of the continuum is {\sc smv}, which by definition
makes no attempt to partition references because it consists of a
single thread. For the SMVP operation $w = Kv$, {\sc lmv} has a better
partition than {\sc smv}, with disjoint references of $K$ and non-disjoint
references of $v$ and $w$.  {\sc Rmv} has a better partition than {\sc lmv}
because it replaces non-disjoint references of $w$ with disjoint ones;
however, references to $v$ are still non-disjoint. Finally, {\sc mmv} and
{\sc hmv} use the best partitions, with disjoint references to $K$, $u$, and
$v$.

The quality of the partition is directly related to the difficulty of
developing and parallelizing the program. For example, the sequential
{\sc smv} program is the simplest program to develop because by
definition it does not partition any references. {\sc Lmv} is only a
little harder to write than {\sc smv}, requiring only simple changes
to {\sc smv}: assigning rows of $K$ to threads, creating the threads,
and protecting updates to the result vector with locks. {\sc Rmv}
requires only simple modifications to {\sc lmv}: defining an array of
vectors and modifying the call to the local SMVP routine. {\sc Mmv}
and {\sc hmv} are much more difficult to develop because they require
a sophisticated partitioner, computation of global-to-local node and
element mappings for each PE, computation of the sparse matrix
structure for each PE, and a communication schedule for each PE.

The quality of the partition is also directly related to
performance. This can be seen quite clearly in
Figure~\ref{fig:sf5.sgi.mflops} for the Spark98 sf5 kernels. The
kernels were run on 1, 2, 4, and 8 PEs of a 12-PE SGI Power Challenge,
a bus based cache coherent shared memory system that supports both
shared memory and MPI message passing primitives.  The {\sc lmv}
program was run using $k = 1, 2, 4, \ldots, 512$ locks, where 512 is
the maximum (power of two) number of locks allowed by the system.  The
performance of the baseline sequential {\sc smv} program 
is shown as a single dot in the lower lefthand corner of the
graph. Performance is reported in absolute throughput units of Mflop/s
(millions of floating point operations per sec).\footnote{We use
Mflop/s rather than MFLOPS, as suggested by \cite{hockney96}.} 

\begin{figure}
\centerline{ 
%\epsfxsize=3in
\epsfbox{sf5.sgi.mflops.epsf} 
}
\caption{Performance of the Spark98 sf5 kernels on an SGI Power Challenge
(195 MHz MIPS R10000, 32 Kbyte Icache, 32 Kbyte L1 Dcache, 2 Mbyte L2
Dcache, cc -O2). Sample points are at 1, 2, 4, and 8 PEs. {\sc Lmv}/sf5 was
measured using 1, 2, 4, 8, 16, 32, 64, 128, 256, and 512 locks. {\sc Smv}/sf5
performance is 39 Mflop/s.}
\label{fig:sf5.sgi.mflops}
\end{figure}

Figure~\ref{fig:sf5.sgi.mflops} shows how performance improves as
programs do a better job of partitioning their data references. {\sc Mmv}
and {\sc hmv} are faster than {\sc rmv}, which is faster than all instances of
{\sc lmv}. Although the single-PE {\sc lmv} kernel is slower than the purely
sequential {\sc smv} program because of lock overheads, given enough locks
and threads its performance quickly surpasses that of {\sc smv}.  However,
for small numbers of locks, {\sc lmv} actually experiences a slowdown as the
number of threads increase, due to excessive contention for locks.
Also, notice how poorly the performance of {\sc rmv} scales, as we predicted
in Section~\ref{sec:programs}.

Conventional wisdom maintains that shared memory programs are easier
to write, but less efficient, than message passing programs. This
statement is certainly true for {\sc lmv} and {\sc rmv}, which are simpler to
write than {\sc mmv} and provide less than half the performance. However,
the statement is not true for {\sc hmv}, which is just as difficult to write
as {\sc mmv} and has equivalent performance.  Thus, on the one hand an
efficient shared memory program is not necessarily easier to write
than a message passing program, but on the other hand it is not
necessarily slower either.
    
Although {\sc mmv} and {\sc hmv} enjoy the best performance,
Figure~\ref{fig:sf5.sgi.mflops} shows that the performance curve is
flattening as the number of PEs increases to 8 PEs, a very small
number of PEs indeed. This type of scalability problem does not arise
in every system, as shown in Figure~\ref{fig:sf5.t3e.mflops} for the
identical {\sc mmv}/sf5 kernel running on the Cray T3E, which uses a torus
instead of a shared bus to connect the PEs. Designers can use the
kernels in this way to help them identify limitations in the
communication systems of their parallel computers.
\begin{figure}
\centerline{ 
%\epsfxsize=3in
\epsfbox{sf5.t3e.mflops.epsf} 
}
\caption{Performance of the Spark98 {\sc smv}/sf5 and {\sc mmv}/sf5 kernels on a Cray T3E
(450 MHz Alpha 21164, 8 Kbyte Icache, 8 Kbyte L1 Dcache, 96 Kbyte L2 Dcache, cc -O2).
Sample points are at 1, 2, 4, and 8 PEs.
{\sc Smv}/sf5 performance is 57 Mflop/s.}
\label{fig:sf5.t3e.mflops}
\end{figure}

Finally, the sequential {\sc smv} program illustrates an interesting
property of modern compilers and processors.  It is not unreasonable
to wonder if more aggressive use of temporary variables in the {\em
local\_smvp()} routine might improve performance by eliminating the
possibility of pointer aliasing. Interestingly enough, though, loading
every input value into a temporary scalar variable makes almost no
difference in performance on a modern system.  For example, on an SGI
Power Challenge, rewriting {\em local\_smvp()} to use as many
temporaries as possible raises the measured performance of the {\sc
smv}/sf5 kernel from 39 Mflop/s to 40.5 Mflop/s, an improvement of
only 4\%.

\section{Concluding remarks}

Our purpose in developing Spark98 is to give system builders access to
parallel sparse matrix kernels that are realistic, and yet fairly
small and easy for one person to understand, modify, and experiment
with. The kernels are based on a baseline sequential SMVP program,
three shared memory programs, and a message passing program, along
with two unstructured three-dimensional meshes from an earthquake
ground modeling application. The programs represent points along a
continuum that is defined by the care with which data references are
partitioned among PEs.

Although we expect that the current set of kernels will be quite
useful, there are some limitations that we might want to address in
the future.  First, as we have already discussed, the Spark98 programs
are points along a continuum defined by how well they partition data
references among their computational threads. We don't claim to have
provided complete coverage of parallel SMVP algorithms, and there are
surely other approaches that would be interesting to add to the mix of
programs.

Second, it might be useful to include other computations beside the
SMVP in the kernels. For example, examples of direct methods 
for solving sparse linear systems would be helpful.

Finally, the current set of kernels are based on two finite element
meshes from the CMU Quake project. It might be useful to include
meshes from other types of applications such as computational fluid
dynamics.

\section*{Acknowledgments}
The Spark98 meshes were created by my colleagues on the CMU Quake
project: Jacobo Bielak, Omar Ghattas, Loukas Kallivokas, Hesheng Bao,
and Jifeng Xu.  The meshes were generated, partitioned, and parceled
using programs from the Archimedes tool chain that were written by
Jonathan Shewchuk while he was a graduate student at Carnegie
Mellon. Jifeng Xu also wrote part of an octree program that generates
the initial distribution of mesh nodes. Thomas Gross gave helpful
perspective on how to develop and organize the kernels. Computer time
on the Cray T3E and SGI Power Challenge was provided by the Pittsburgh
Supercomputing Center.

\bibliography{/afs/cs/project/iwarp/member/droh/bib/refs}

\end{document}


























